Author’s response to reviews

Title: Clostridium perfringens panophthalmitis and orbital cellulitis: a case report

Authors:

Ghita Ghedira (rita_guedira@hotmail.com)
Nabil Taright (ntaright@gmail.com)
Hélène Blin (Blin.Helene@chu-amiens.fr)
Thameur Fattoum (t.fattoum@ch-beauvais.fr)
Jordan Leroy (leroy.jordan80@gmail.com)
Youssef El Samad (elsamad.youssef@chu-amiens.fr)
Solange Millazo (millazo.solange@chu-amiens.fr)
Farida Hamdad (hamdadjrada2002@yahoo.fr)

Version: 2 Date: 31 Dec 2017

Author’s response to reviews:

Dear Editor, Dear reviewers,

We thank you for your interest in our article

In patients with penetrating eye injuries from contaminated foreign bodies, the physicians must aware that penetrating injury acquired in rural setting is more likely to progress to endophtalmitis than are penetrating injuries in other setting. On the other hand, more virulent pathogens as C. perfringens are likely to be encountered.

As described by Kelly et al. the best management was early vitrectomy and their patient had the best outcome that ours.

There have been errors in the management of our patient and this is precisely what we must avoid in the future.

Some corrections were performed to the article and a figure of the prosthesis was added. All corrections have been highlighted.

Thank you for all and Happy New Year.